vc/9-May-03
Week 19: (12/05 to 15/05)

SM18 Operation SCHEDULE (Monday Morning) Meeting 

· Test Coordinator this week:    O.Berrig

· MTA Technician on call:  (16-3602) Richard.

1. Planning for this week : 
· TBA1:   3007  :  Tests after Final warm up .1016 is coming (prepa. at warm)
· TBA2:   2006  Test at cold after thermal cycle (+tests non performed during first run).
· TBC2 :  3013 at 1.9 K (still first run) 
·  TBC1 : 1024 Ready to be cooled down .
· *** CONTINUE  USAGE of “ To-do list ”    *** 
NEW LIST (MASTER COPY) ON PUT On BOARD FOR Thermal Cycle Type Magnets 
2.  Follow-up of problems from Operation Last week( Marco+ Richard)
3. NEWS (if any) From Cryogenics:  J-P Lamboy
Problem with cold and HP compressor: news?
4. Other General Issues

· Shift hours for operators :    Morning 6h00-13h00
Afternoon : 13h00-21h00
Night 21h00-6h00.

· Connection/disconnection current leads: Training organized by A.Siemko. (names of volunteers)

